Author's response to reviews

Title: Quality assessment of DNA derived from up to 30 years old formalin fixed paraffin embedded tissue (FFPE) for PCR-based methylation analysis using SMART-MSP and MS-HRM

Authors:

Lasse S Kristensen (lasse@humgen.au.dk)
Tomasz K Wojdacz (wojdacz@humgen.au.dk)
Britta B Thstrup (boserup@humgen.au.dk)
Carsten Wiuf (wiuf@birc.au.dk)
Henrik Hager (henrikhager@mac.com)
Lise Lotte Hansen (Lotte@humgen.au.dk)

Version: 4 Date: 27 November 2009

Author's response to reviews:

The authors competing interests section is included in the revised manuscript.

The entire manuscript is formatted according to the journal template for BMC medicine journals as recommended.